Clinical research often takes place in developing countries. The ethical issues related to research although important worldwide are particularly sensitive in developing countries, therefore, enforcement of Informed Consents has been a major goal for the bioethicist. When following the international rules to consent the individuals; often, social, cultural and economical factors may not be considered, as a result the purpose of informed consent is not achieved. In disease like cancer the clinical trials are important part of treatment and making sure that a valid consent is taken is important.

In this work I try to probe that in the actual setting the consent procedure is not the ethical panacea for patients living in developing countries are in which it is important to enforce that the trials h